Author's response to reviews

Title: Fracture prevention with vitamin D supplementation: considering the inconsistent results

Authors:

Gerbrand J Izaks (g.j.izaks@int.umcg.nl)

Version: 5 Date: 9 March 2007

Author's response to reviews:

Dear Sir, Madam,

As required in your e-mail of March 8, the following changes were made:

- website address was removed from the text and added to the main reference list.
- competing interests: text changed according to your requirements.
- acknowledgements: I am the sole author of this paper. There are no other contributors who should be acknowledged.
- figure legends: text changed according to your requirements.

Best regards,

Gerbrand Izaks
Dept. General Internal Medicine and Geriatrics
University Medical Centre Groningen
The Netherlands